Author's response to reviews

Title: Perinatal HIV transmission and the cost-effectiveness of screening at 14 weeks gestation, at the onset of labour and the rapid testing of infants.

Authors:

Belinda Udeh (belinda-udeh@uiowa.edu)
Chiedozie Udeh (n.graves@qut.edu.au)
Nicholas Graves (n.graves@qut.edu.au)

Version: 4 Date: 18 December 2008

Author's response to reviews:

Dear Dr. Norton,

Thank you for your latest email about this paper.

We have made all the changes required but could not find any inconsistency with the numbering of referencing. References 28, 29, 45 appeared to us to follow in the correct order. Please let us know if we are wrong on this.

Thank you again for the time and effort by yourself, editorial board and reviewers and thank you for the acceptance of our submission for publication.

On behalf of the authors,

Nicholas Graves, PhD